Author’s response to reviews

Title: Single center experience with laparoscopic adrenalectomy on a large clinical series.

Authors:
Giovanni Conzo (giovanni.conzo@unicampania.it)
Claudio Gambardella (claudiog86@hotmail.it)
Giancarlo Candela (giancarlo.candela@unicampania.it)
Alessandro Sanguinetti (sanguinettiale@gmail.com)
Andrea Polistena (apolis74@yahoo.it)
Guglielmo Clarizia (clarizia.guglielmo@libero.it)
Renato Patrone (dott.patrone@gmail.com)
Francesco Di Capua (francesc.dicapua@gmail.com)
Chiara Offi (chiara.o@live.com)
Sergio Iorio (sergio.iorio@unicampania.it)
Giuseppe Bellastella (giuseppe.bellastella@unicampania.it)
Daniela Pasquali (daniela.pasquali@unicampania.it)
Annamaria De Bellis (annamaria.debellis@unicampania.it)
Antonio Sinisi (antonio.sinisi@unicampania.it)
Nicola Avenia (nicola.avenia@unicampania.it)

Version: 3 Date: 20 Nov 2017

Author’s response to reviews:

Dear edithor, thank you for revising our manuscript and giving us good advices in order to improve it. Here we respond about the changes you asked:

1) the last paragraph of results was modified in order to evitate overlaps

2) all the authors filled the change-in-autorship form and it was sent to editorial office
3-4-5-6) we applied the changes you asked

7) the ethical approval was not asked anymore as the study was retrospective and we only collected datas, so the ethical approval was unnecessary. We modified the wording in "not applicable"

8) initials were changed

9) full names were inserted

10) tracked changes and highlightings were removed

Hoping to have fullfilled all your requests in the right way, we look forward to recieve your respones

Kind regard,

Giovanni Conzo